Author's response to reviews

Title: Essential Pediatric Hypertension: Defining the Educational Needs of Primary Care Pediatricians

Authors:

Stephen D Cha (scha@chmca.org)
Deena J Chisolm (Deena.Chisolm@nationwidechildrens.org)
John D Mahan (John.Mahan@nationwidechildrens.org)

Version: 5 Date: 11 July 2014

Author's response to reviews: see over
Dear editors,

We are submitting a revised version of our qualitative research article entitled “Essential Pediatric Hypertension: Defining the Educational Needs of Primary Care Pediatricians”, by Stephen Cha et al to be considered for publication as a research article in BMC Medical Education. All co-authors have seen and approve of the contents of the manuscript and there are no financial interests to disclose. We certify that the submission is original work and is not under review at any other publication.

We have added a new version of the manuscript with all changes from the previous submission highlighted in yellow. We believe the revised manuscript addresses all of the editor’s concerns.

1. We addressed the question regarding formal consent and added a statement into the manuscript that states that verbal consent was obtained.
2. We have reviewed the RATS guideline for qualitative research and our methods and analysis meets these requirements.

We hope that the editorial board and reviewers will agree on the interest of this manuscript and accept our modification. Thank you for considering this article for submission.

Sincerely yours,

Stephen Cha, MD (on behalf of the authors)
Pediatric Nephrology Fellow
Nationwide Children’s Hospital